Author's response to reviews

Title: Analysis on the clinical features of 22 basaloid squamous cell carcinoma of the lung

Authors:

Li C Wang (wangchangli123@gmail.com)
Lei Wang (wanglei345@gmail.com)
Sabrina Kwauk (skwauk@gmail.com)
Jennifer A Woo (jennifer.woo@gmail.com)
Li Q Wu (wugiuli243@gmail.com)
Hong Zhu (zhuhong@tijmu.edu.cn)
Li Z Zhan (zhanzhungli3@gmail.com)
Na L Sun (sunleina77@gmail.com)
Lei Zhang (raymd728@yahoo.com.cn)

Version: 5 Date: 23 January 2011

Author's response to reviews:

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript and the list of authors in the submission system should be identical, both in style and order. The preferred style is: First name Initial Last name (e.g. Joe F Bloggs)

It has been done.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example: The Title of My Prestigious Publication: A Scientific Study. should be changed to: The title of my prestigious publication: a scientific study

It has been done.

Emails: please use authors' initials in the email list and not full names.

It has been done.

Abbreviations: Please format these in to a paragraph in alphabetical order.

It has been done.

Figures: It is important for the final layout of the manuscript that the figures are
cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures
It has been done.

References: Please make sure that references are cited in the text by the corresponding number from the reference list and not just by author name/year. 'Moro et al. in 2008' on page 6.
It has been done.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
It has been done.